

# Turbulence in the Lower Atmosphere of Mars Enhanced by Transported Dust Particles

Aymeric Spiga

## ▶ To cite this version:

Aymeric Spiga. Turbulence in the Lower Atmosphere of Mars Enhanced by Transported Dust Particles. Journal of Geophysical Research. Planets, 2021, 126, 10.1029/2021JE007066. insu-03726933

# HAL Id: insu-03726933 https://insu.hal.science/insu-03726933

Submitted on 28 Jul2022

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

Copyright



# **JGR** Planets

## **COMMENTARY**

10.1029/2021JE007066

This article is a comment on Wu et al. (2021), https://doi.org/10.1029/2020JE 006752.

#### **Key Points:**

- A recent study reports unprecedented turbulence-resolving numerical simulations for Mars
- A dustier Martian atmosphere may result in stronger turbulence in Mars' planetary boundary layer
- This has many implications for future studies of Mars' meteorology and climatology

#### Correspondence to:

A. Spiga, aymeric.spiga@sorbonne-universite.fr

#### **Citation**:

Spiga, A. (2021). Turbulence in the lower atmosphere of Mars enhanced by transported dust particles. *Journal* of *Geophysical Research: Planets*, 126, e2021JE007066. https://doi. org/10.1029/2021JE007066

Received 21 SEP 2021 Accepted 1 NOV 2021

# Turbulence in the Lower Atmosphere of Mars Enhanced by Transported Dust Particles

## Aymeric Spiga<sup>1,2</sup>

<sup>1</sup>Laboratoire de Météorologie Dynamique/Institut Pierre-Simon Laplace (LMD/IPSL), Sorbonne Université, Centre National de la Recherche Scientifique (CNRS), École Polytechnique, École Normale Supérieure (ENS), Campus Pierre et Marie Curie, Paris, France, <sup>2</sup>Institut Universitaire de France (IUF), Paris, France

**Abstract** A recent article by Wu et al. (2021, https://doi.org/10.1029/2020JE006752) reports unprecedented turbulence-resolving numerical simulations for Mars, in which the radiative impact of suspended dust particles transported in convective circulations is resolved for the first time. Those simulations demonstrate that in certain conditions, a dustier Martian atmosphere may result in stronger turbulence and mixing in Mars' planetary boundary layer, as a result of the inhomogeneity in the distribution of dust particles being reinforced by turbulent convective cells. This effect competes with what was hitherto thought to dominate in a dustier atmosphere, that is, a turbulence weakening related to surface shading by suspended dust particles. Wu et al. (2021 https://doi.org/10.1029/2020JE006752)'s work has many implications for future studies of Mars' meteorology and climatology.

**Plain Language Summary** Near the surface of Mars, the air is strongly turbulent in daytime as a result of the warm surface heated by sunlight. Before Wu et al. (2021, https://doi.org/10.1029/2020JE006752)'s study, more dust in Mars' air was thought to lead to weaker turbulence because the added dust particles shade the surface. On the contrary, using unprecedented computer simulations calculating the turbulent motions of Martian near-surface air and of the dust particles it carries, Wu et al. (2021, https://doi.org/10.1029/2020JE006752) showed that more dust in Mars' air could lead to *stronger* turbulence because dust particles locally warm the upward turbulent plumes in which they concentrate. This new result changes the way scientists may view Mars' weather and climate in the future.

Dust is for weather on Mars what water vapor is for weather on Earth: dust particles suspended in the Martian atmosphere are a key source of meteorological and climatological variability. It is not entirely out of chance that terms inherited from terrestrial moist convection studies are borrowed by Martian scientists (Heavens et al., 2019) to designate observed or modeled dust storms on Mars: for example, hurricanes (Rafkin, 2009), cumulonimbus (Spiga et al., 2013), and mesoscale textures (Guzewich et al., 2015; Heavens, 2017). Airborne dust particles suspended in the Martian atmosphere are crucial ingredients of radiative forcing on Mars, given their absorption of incoming sunlight in visible wavelengths as well as their contribution to radiative fluxes in thermal infrared wavelengths (Kahre et al., 2017; Määttänen & Savijärvi, 2004; Madeleine et al., 2011). Local-to-regional dust storms on Mars (Battalio & Wang, 2021; Cantor, 2007; Määttänen et al., 2009) are the main source of interannual variability and their triggering, growth, and decay are still an active topic of research given the complex radiative-dynamical feedbacks that govern their evolution.

How dust particles get lifted from Mars' surface then suspended in Mars' atmosphere depends critically on winds and circulations developing in the atmospheric layers a couple of kilometers above the surface, the so-called planetary boundary layer (see, for Mars, Petrosyan et al., 2011 for an expert-level review, and Spiga, 2019 for a review aimed at non-expert audience). Mars is an environment known since the Viking era (Sutton et al., 1978) for its very active turbulence in the planetary boundary layer, especially in the daytime. This strong turbulence activity has been experienced by all the in-situ spacecraft that touched down the surface of Mars (see e.g., the review of in-situ observations by Martínez et al., 2017) and its impact on the thermal structure and volatile mixing have been detected by many orbital spacecraft (Fenton et al., 2016; Hinson et al., 2008; Malin & Edgett, 2001; Stanzel et al., 2008). For readers not expert of Martian weather, it might be deemed surprising that the cold Martian environment would be prone to such a strong near-surface atmospheric turbulence. Nevertheless, more than the absolute temperature, what matters in driving

© 2021. American Geophysical Union All Rights Reserved.

atmospheric turbulence in the near-surface environment is the contrast between surface and atmospheric temperature and this contrast is particularly strong in the thin Martian atmosphere. The daytime turbulence on Mars is actually even stronger than expected solely from this temperature contrast, because of near-surface infrared absorption of incoming surface radiation by  $CO_2$  (the main component of the Martian atmosphere) and, to lesser extent, by dust particles (Haberle et al., 1993; Savijärvi, 1999; Spiga et al., 2010). In other words, the Martian planetary boundary layer is much more radiatively controlled than its terrestrial counterpart (Spiga, 2011).

The impact of airborne dust particles on atmospheric turbulence is thought to be fairly straightforward in most existing studies. Owing to absorption by suspended dust particles, dusty conditions on Mars are expected to cause a shading of incoming sunlight at the surface, hence a decrease of the strength of turbulence in the planetary boundary layer, and a subsequent negative feedback for dust lifting (Kahre et al., 2006; Newman et al., 2002). Nevertheless, a handful of studies argues for a possible enhancement of convective structures in the planetary boundary layer by transported dust particles (Daerden et al., 2015; Fuerstenau, 2006; Heavens et al., 2011; Rafkin, 2012; Spiga et al., 2013). No existing study proposes, however, the kind of advanced numerical simulations that Wu et al. (2021) carried out to advance significantly the knowledge on the impact of suspended dust particles on Mars' convective turbulence in the planetary boundary layer.

The numerical modeling approach that Wu et al. (2021) used is named large-eddy simulations. This technique is used on Earth to explore turbulence in the planetary boundary layer: large-eddy simulations are fine-scale simulations in which only the largest turbulent eddies a couple of tens of meters across, responsible for most of the mixing in the planetary boundary layer, are explicitly resolved (Couvreux et al., 2010; Giersch et al., 2019; Klose & Shao, 2013; Lilly, 1962; Mason, 1989). Large-eddy simulations provide fruitful insights in the case of the Martian atmosphere too (Odaka et al., 1998; Rafkin et al., 2001), to understand convective vortices (Kanak, 2006; Nishizawa et al., 2016; Toigo et al., 2003) leading to dust devils (Lorenz et al., 2016; Spiga et al., 2016), to assess the horizontal and vertical structure of the convective planetary boundary layer (Fenton & Michaels, 2010; Michaels & Rafkin, 2004; Richardson et al., 2007; Sorbjan, 2007; Spiga et al., 2010), to determine the atmospheric dynamics encountered by landers and rovers during their Entry, Descent, and Landing (Rafkin & Michaels, 2003; Tyler et al., 2008), to make sense of turbulence observations by spacecraft visiting Mars (Davy et al., 2010; Gheynani & Taylor, 2011; Spiga et al., 2021), and to explore the possibility of radiatively controlled turbulent convection in Mars' nighttime water-ice clouds (Spiga et al., 2017).

The study of Wu et al. (2021) proposes, for the first time, large-eddy simulations of the dusty Martian atmosphere where dust particles are not being simply used to compute radiative forcings in the atmosphere, but are also actively transported in the turbulent motions resolved by their model. What Wu et al. (2021) basically do is to introduce an additional level of complexity (i.e., explicitly resolving the dust-related radiative-dynamical feedbacks related to turbulence on Mars) to reach a more realistic representation of Mars' dusty atmosphere in turbulence-resolving models and to conduct unprecedented numerical experiments, akin to laboratory experiments with brand new instruments. This is obviously a risky strategy, involving time-consuming model development and validation, but possibly a highly fruitful approach.

The new simulation pathway proposed by Wu et al. (2021)—coupling turbulence-resolving simulations to radiatively active transported dust particles–allows them to discover that, contrary to what has been thought thus far, a dustier Martian atmosphere could be, in certain conditions, *more* turbulent than a clear Martian atmosphere (Figure 1). A preliminary to those new results is first to check that, in known case studies, their model used for large-eddy simulations are consistent with existing spacecraft observations (Hinson et al., 2008) and previously published large-eddy simulations (Spiga et al., 2010). Wu et al. (2021) also confirm that the configuration adopted by existing large-eddy simulations, that is, a fixed dust distribution, yields the result previously considered as the consensus—that is, an increase in atmospheric dust opacity causes a weakening of turbulent convection owing to reduced sunlight reaching the surface. After this thorough validation, Wu et al. (2021) report their new, unprecedented large-eddy simulations using interactive transport of dust particles that show that, contrary to the fixed distribution case, convection may be stronger in dustier cases. This results from the horizontal structure of turbulent circulations (in particular cellular structures akin to Rayleigh-Bénard convection cells) causing an inhomogeneous spatial distribution of dust particles, which are primarily concentrating in updrafts (upwelling convection plumes). Consequently, the







**Figure 1.** Schematics describing the interplay in Mars' atmosphere between daytime turbulence in the planetary boundary layer (black arrows) and the amount of suspended dust particles (orange dots). The top panel shows the regular situation where the atmosphere of Mars is not very dusty: the surface is warmed by incoming sunlight and turbulence is active. The middle and bottom panels show the case when/where the atmosphere of Mars is dustier than usual. In the middle panel, according to previous studies, more suspended dust particles means enhanced surface shading compared to the top panel situation (because part of the sunlight is absorbed by dust particles). As a result, turbulence on Mars is systematically weaker in a dustier atmosphere than in a clear atmosphere, with a lower mixing height from the surface. In the bottom panel, according to the new study by Wu et al. (2021), dust particles are transported by turbulent convective cells and concentrate within upward plumes where they warm locally the atmosphere. Consequently, more suspended dust particles mean (a) stronger horizontal temperature differences and (b) stronger buoyancy. As a result, contrary to what was previously thought, turbulence on Mars could be stronger in a dustier atmosphere than in a clear atmosphere, with a larger mixing height from the surface meaning that dust particles and chemical species could be transported higher than previously thought. The situation described in the middle panel remains valid when the atmosphere of Mars is homogeneously very dusty.

absorption of incoming sunlight by dust particles is enhanced in turbulent updrafts, which has two effects (see Figure 1) simultaneously acting to reinforce convective motions: (1) in the horizontal dimension, thermal contrasts in the CBL are stronger and (2) in the vertical dimension, buoyancy production is larger. In other words, in addition to the already-known stabilizing effect of atmospheric dust loading (by shading of the surface), Wu et al. (2021) demonstrate with their new modeling approach that there is a competing destabilizing effect of atmospheric dust loading related to the concentration of dust particles in upwelling plumes. Wu et al. (2021) identify that only certain combinations of atmospheric dust loading (total optical depth) and spatial dust inhomogeneity—for instance a moderate increase in dust loading with active areas of dust lifting—leads to the destabilizing effect to dominate the stabilizing effect, thereby causing the dustier atmosphere to exhibit stronger turbulence in the planetary boundary layer; in other circumstances, in particular, when the atmosphere is homogeneously very dusty, the stabilizing effect dominates and this corresponds to the regime known before Wu et al. (2021)'s study.

The implications of Wu et al. (2021)'s work are broad and diverse. An example of possibly crucial implication is related to the growth of dust storms on Mars. The possible positive feedback between atmospheric dust loading and the vigor of turbulence in the daytime planetary boundary layer is interesting, because a more vigorous turbulence means more powerful wind gusts, henceforth enhanced lifting, injection, and



mixing of dust particles from the surface to the overlying atmosphere. This is a possible mechanism to explain the growth of a local dust storm toward a larger regional dust storm. The fact that this positive feedback may only exist in a certain range of atmospheric dust loading and dust inhomogeneity is an additional interesting perspective to explain why only certain local dust storms grow into larger dust storms while others do not (Toigo et al., 2018). Another possible great implication of this work is the exploration of various mechanisms for regional and seasonal variability of the strength of mixing in Mars' planetary boundary layer. Dust loading now comes—and in a subtle way, since this depends on both total dust loading and dust inhomogeneity—in addition to incoming sunlight, topography (Hinson et al., 2008; Spiga et al., 2010), and slope winds (Hinson et al., 2019) as potential causes for spatial and seasonal variability of mixing in the near-surface atmosphere. This could be key to understand how aerosols, volatiles, and chemical species are mixed and transported in the Martian atmosphere, from the near-surface to the upper atmosphere. The results of Wu et al. (2021) could imply that in a season and/or region with a dustier atmosphere, the vertical transport and mixing of species from the surface to the upper atmosphere is underestimated in global circulation models not resolving the intimate coupling between dust particles and turbulent motions. Thus, a related outcome of the turbulence-resolving simulations by Wu et al. (2021) is that the parameterizations of subgrid-scale mixing by turbulence in global climate models (e.g., Colaïtis et al., 2013) will need to be upgraded to account for those new effects. A last implication is a fact that the future robotic and human exploration of Mars calls for an improved understanding of Martian weather to reach a better level of predictability. Getting a view on new potential feedback mechanisms as exhibited in Wu et al. (2021) for the crucial dust-related processes on Mars allows the Martian community to be in a better place to fulfill this goal.

An obvious future line of research is that, as both thought-provoking and convincing as those new results of Wu et al. (2021) are, these are based on modeling and need to be validated by observations in future work. This constructive dialogue between observations and modeling that underlies the contemporary Martian atmospheric science is the best proof of its maturity and also the best pathway for future progress. The confirmation of the modeling results by Wu et al. (2021) may lie in existing in-situ and orbital data set already acquired on Mars, or could require additional measurement campaigns. For instance, a fascinating possibility is that the mechanism depicted by Wu et al. (2021) could explain the observed strong enhancement of vortex-induced pressure drops during the "dusty" season by both Curiosity (Ordonez-Etxeberria et al., 2018; Newman et al., 2019) and InSight (Chatain et al., 2021). On this topic, the future in situ atmospheric measurements by Perseverance and Zhurong, at different locations than Curiosity and InSight, will be particularly interesting. Another great perspective to build on Wu et al. (2021)'s modeling work is to explore whether an enhanced planetary boundary layer mixing in dusty conditions could be evidenced in existing (or future) temperature profiling acquired by thermal infrared spectrometers and radio occultations. At any event, future innovative measurements will also be needed to better understand Mars' planetary boundary layer, the part of the atmosphere crucial to both human/robotic exploration and surface-atmosphere interactions. The modeling study by Wu et al. (2021) underlines the need for a significant leap in the techniques deployed to profile Mars' planetary boundary layer, for instance by combining instrumentation never sent to Mars-like sonic anemometers, wind/aerosols LIDAR, dust flux experiments-on board a new generation of Martian spacecraft, for example, a global network of fixed weather stations, associated with instrumented aerial explorers building on the successful demonstration flights of NASA's Ingenuity helicopter. The experience coming from terrestrial experts on this endeavor will be particularly crucial.

## **Data Availability Statement**

Data were not used, nor created for this research.

### References

- Battalio, M., & Wang, H. (2021). The Mars Dust Activity Database (MDAD): A comprehensive statistical study of dust storm sequences. *Icarus*, 354, 114059. https://doi.org/10.1016/j.icarus.2020.114059
- Cantor, B. A. (2007). MOC observations of the 2001 Mars planet-encircling dust storm. *Icarus*, 186, 60–96. https://doi.org/10.1016/j. icarus.2006.08.019
- Chatain, A., Spiga, A., Banfield, D., Forget, F., Murdoch, N., Chatain, A., et al. (2021). Seasonal variability of the daytime and nighttime atmospheric turbulence experienced by InSight on Mars. *Geophysical Research Letters*. https://doi.org/10.1029/2021gl095453

#### Acknowledgments

Aymeric Spiga thanks the JGR-planets editors for the invitation to write this Commentary and Associate Editor Germán Martínez for thoughtful comments.



- Colaïtis, A., Spiga, A., Hourdin, F., Rio, C., Forget, F., & Millour, E. (2013). A thermal plume model for the Martian convective boundary layer. *Journal of Geophysical Research (Planets)*, *118*, 1468–1487. https://doi.org/10.1002/jgre.20104
- Couvreux, F., Hourdin, F., & Rio, C. (2010). Resolved versus parametrized boundary-layer plumes. Part I: A parametrization-oriented conditional sampling in large-eddy simulations. *Boundary-Layer Meteorology*, 134, 441–458. https://doi.org/10.1007/s10546-009-9456-5 Daerden, F., Whiteway, J. A., Neary, L., Komguem, L., Lemmon, M. T., Heavens, N. G., et al. (2015). A solar escalator on Mars: Self-lifting
- of dust layers by radiative heating. Geophysical Research Letters, 42(18), 7319–7326. https://doi.org/10.1002/2015GL064892
- Davy, R., Davis, J. A., Taylor, P. A., Lange, C. F., Weng, W., Whiteway, J., & Gunnlaugson, H. P. (2010). Initial analysis of air temperature and related data from the Phoenix MET station and their use in estimating turbulent heat fluxes. *Journal of Geophysical Research (Planets)*, 115(E14), E00E13. https://doi.org/10.1029/2009JE003444
- Fenton, L., & Michaels, T. I. (2010). Characterizing the sensitivity of daytime turbulent activity on Mars with the MRAMS LES: Early results. *International Journal of Mars Science and Exploration*, 5, 159–171. https://doi.org/10.1555/mars.2010.0007
- Fenton, L., Reiss, D., Lemmon, M., Marticorena, B., Lewis, S., & Cantor, B. (2016). Orbital observations of dust lofted by daytime convective turbulence. Space Science Reviews, 203(1), 89–142. https://doi.org/10.1007/s11214-016-0243-6
- Fuerstenau, S. D. (2006). Solar heating of suspended particles and the dynamics of Martian dust devils. *Geophysical Research Letters*, 33, 19. https://doi.org/10.1029/2006GL026798
- Gheynani, B. T., & Taylor, P. A. (2011). Large eddy simulation of typical dust devil-like vortices in highly convective Martian boundary layers at the Phoenix lander site. *Planetary and Space Science*, 59, 43–50. https://doi.org/10.1016/j.pss.2010.10.011
- Giersch, S., Brast, M., Hoffmann, F., & Raasch, S. (2019). Toward large-eddy simulations of dust devils of observed intensity: Effects of grid spacing, background wind, and surface heterogeneities. *Journal of Geophysical Research: Atmospheres*, 124(14), 7697–7718. https://doi. org/10.1029/2019jd030513
- Guzewich, S. D., Toigo, A. D., Kulowski, L., & Wang, H. (2015). Mars Orbiter Camera climatology of textured dust storms. *Icarus*, 258, 1–13. https://doi.org/10.1016/j.icarus.2015.06.023
- Haberle, R. M., Pollack, J. B., Barnes, J. R., Zurek, R. W., Leovy, C. B., Murphy, J. R., et al. (1993). Mars atmospheric dynamics as simulated by the NASA/Ames general circulation model, 1, the zonal-mean circulation. *Journal of Geophysical Research*, *98*(E2), 3093–3123. https://doi.org/10.1029/92je02946
- Heavens, N. G. (2017). Textured dust storm activity in Northeast Amazonis-Southwest Arcadia, Mars: Phenomenology and dynamical interpretation. Journal of the Atmospheric Sciences, 74(4), 1011–1037. https://doi.org/10.1175/JAS-D-16-0211.1
- Heavens, N. G., Kass, D. M., Shirley, J. H., Piqueux, S., & Cantor, B. A. (2019). An observational overview of dusty deep convection in Martian dust storms. Journal of the Atmospheric Sciences, 76(11), 3299–3326. https://doi.org/10.1175/JAS-D-19-0042.1
- Heavens, N. G., Richardson, M. I., Kleinböhl, A., Kass, D. M., McCleese, D. J., Abdou, W., et al. (2011). Vertical distribution of dust in the Martian atmosphere during northern spring and summer: High-altitude tropical dust maximum at northern summer solstice. *Journal* of Geophysical Research, 116(E15), E01007. https://doi.org/10.1029/2010JE003692
- Hinson, D. P., Pätzold, M., Tellmann, S., Häusler, B., & Tyler, G. L. (2008). The depth of the convective boundary layer on Mars. *Icarus*, 198, 57–66. https://doi.org/10.1016/j.icarus.2008.07.003
- Hinson, D. P., Tyler, D., Lewis, S. R., Pätzold, M., Tellmann, S., Häusler, B., & Tyler, G. L. (2019). The Martian daytime convective boundary layer: Results from radio occultation measurements and a mesoscale model. *Icarus*, 326, 105–122. https://doi.org/10.1016/j. icarus.2019.02.028
- Kahre, M. A., Murphy, J. R., & Haberle, R. M. (2006). Modeling the Martian dust cycle and surface dust reservoirs with the NASA Ames general circulation model. *Journal of Geophysical Research*, 111(E10), 6008. https://doi.org/10.1029/2005JE002588
- Kahre, M. A., Murphy, J. R., Newman, C. E., Wilson, R. J., Cantor, B. A., Lemmon, M. T., & Wolff, M. J. (2017). The Mars dust cycle. In R. M. Haberle, R. T. Clancy, F. Forget, M. D. Smith, & R. W. Zurek (Eds.), *The atmosphere and climate of mars* (pp. 229229–337294). Cambridge University Press. https://doi.org/10.1017/9781139060172.010
- Kanak, K. M. (2006). On the numerical simulation of dust devil-like vortices in terrestrial and Martian convective boundary layers. Geophysical Research Letters, 33, 19. https://doi.org/10.1029/2006GL026207
- Klose, M., & Shao, Y. (2013). Large-eddy simulation of turbulent dust emission. Aeolian Research, 8, 49–58. https://doi.org/10.1016/j. aeolia.2012.10.010
- Lilly, D. K. (1962). On the numerical simulation of buoyant convection. *Tellus*, *14*(2), 148–172. https://doi.org/10.3402/tellusa.v14i2.9537 Lorenz, R. D., Balme, M. R., Gu, Z., Kahanpää, H., Klose, M., Kurgansky, M. V., et al. (2016). History and applications of dust devil studies.
- Space Science Reviews, 203, 5–37. https://doi.org/10.1007/s11214-016-0239-2 Määttänen, A., Fouchet, T., Forni, O., Melchiorri, R., Forget, F., Savijarvi, H., et al. (2009). A study of the properties of a local dust storm
- with Mars Express OMEGA and PFS data. *Icarus*, 201(2), 504–516. https://doi.org/10.1016/j.icarus.2009.01.024 Määttänen, A., & Savijärvi, H. (2004). Sensitivity tests with a one-dimensional boundary-layer mars model. *Boundary-Layer Meteorology*, 113(3), 305–320. https://doi.org/10.1007/s10546-004-5274-y
- Madeleine, J.-B., Forget, F., Millour, E., Montabone, L., & Wolff, M. J. (2011). Revisiting the radiative impact of dust on Mars using the LMD Global Climate Model. Journal of Geophysical Research (Planets), 116, 11010. https://doi.org/10.1029/2011JE003855
- Malin, M. C., & Edgett, K. S. (2001). Mars Global Surveyor Mars Orbiter Camera: Interplanetary cruise through primary mission. Journal of Geophysical Research, 106, 23429–23570. https://doi.org/10.1029/2000JE001455
- Martínez, G. M., Newman, C. N., De Vicente-Retortillo, A., Fischer, E., Renno, N. O., Richardson, M. I., et al. (2017). The modern near-surface martian climate: A review of in-situ meteorological data from viking to curiosity. Space Science Reviews, 212(1), 295–338. https:// doi.org/10.1007/s11214-017-0360-x
- Mason, P. (1989). Large-eddy simulation of the convective atmospheric boundary layer. *Journal of the Atmospheric Sciences*, 46(11), 1492–1516. https://doi.org/10.1175/1520-0469(1989)046<1492:lesotc>2.0.co;2
- Michaels, T. I., & Rafkin, S. C. R. (2004). Large eddy simulation of atmospheric convection on Mars. Quarterly Journal of the Royal Meteorological Society, 130, 1251–1274. https://doi.org/10.1256/qj.02.169
- Newman, C. E., Kahanpää, H., Richardson, M. I., Martínez, G. M., Vicente-Retortillo, A., & Lemmon, M. T. (2019). MarsWRF convective vortex and dust devil predictions for gale crater over 3 Mars years and comparison with MSL-REMS observations. *Journal of Geophysical Research (Planets)*, 124(12), 3442–3468. https://doi.org/10.1029/2019JE006082
- Newman, C. E., Lewis, S. R., Read, P. L., & Forget, F. (2002). Modeling the Martian dust cycle, 1. Representations of dust transport processes. Journal of Geophysical Research (Planets), 107, 6-1–6-18. https://doi.org/10.1029/2002je001910
- Nishizawa, S., Odaka, M., Takahashi, Y. O., Sugiyama, K.-i., Nakajima, K., Ishiwatari, M., et al. (2016). Martian dust devil statistics from high-resolution large-eddy simulations. *Geophysical Research Letters*, 43(9), 4180–4188. https://doi.org/10.1002/2016GL068896



- Odaka, M., Nakajima, K., Takehiro, S., Ishiwatari, M., & Hayashi, Y. (1998). A numerical study of the Martian atmospheric convection with a two-dimensional anelastic model. *Earth Planets, and Space, 50,* 431–437. https://doi.org/10.1186/bf03352130
- Ordonez-Etxeberria, I., Hueso, R., & Sánchez-Lavega, A. (2018). A systematic search of sudden pressure drops on Gale crater during two Martian years derived from MSL/REMS data. *Icarus*, 299, 308–330. https://doi.org/10.1016/j.icarus.2017.07.032
  - Petrosyan, A., Galperin, B., Larsen, S. E., Lewis, S. R., Määttänen, A., Read, P. L., et al. (2011). The Martian atmospheric boundary layer. *Reviews of Geophysics*, 49, 3005. https://doi.org/10.1029/2010RG000351
  - Rafkin, S. C. R. (2009). A positive radiative-dynamic feedback mechanism for the maintenance and growth of Martian dust storms. Journal of Geophysical Research (Planets), 114(E13), 1009. https://doi.org/10.1029/2008JE003217
  - Rafkin, S. C. R. (2012). The potential importance of non-local, deep transport on the energetics, momentum, chemistry, and aerosol distributions in the atmospheres of Earth, Mars, and Titan. *Planetary and Space Science*, 60, 147–154. https://doi.org/10.1016/j. pss.2011.07.015
  - Rafkin, S. C. R., Haberle, R. M., & Michaels, T. I. (2001). The Mars Regional Atmospheric Modeling System: Model description and selected simulations. *Icarus*, 151, 228–256. https://doi.org/10.1006/icar.2001.6605
  - Rafkin, S. C. R., & Michaels, T. I. (2003). Meteorological predictions for 2003 Mars Exploration Rover high-priority landing sites. Journal of Geophysical Research (Planets), 108(E12), 8091. https://doi.org/10.1029/2002JE002027
- Richardson, M. I., Toigo, A. D., & Newman, C. E. (2007). PlanetWRF: A general purpose, local to global numerical model for planetary atmospheric and climate dynamics. *Journal of Geophysical Research*, 112, E09001. https://doi.org/10.1029/2006je002825
- Savijärvi, H. (1999). A model study of the atmospheric boundary layer in the Mars Pathfinder lander conditions. *Quarterly Journal of the Royal Meteorological Society*, 125(554), 483–493. https://doi.org/10.1002/qj.49712555406
- Sorbjan, Z. (2007). Statistics of shallow convection on Mars based on large-eddy simulations. Part 1: Shearless conditions. Boundary-Layer Meteorology, 123, 121–142. https://doi.org/10.1007/s10546-006-9128-7
- Spiga, A. (2011). Elements of comparison between Martian and terrestrial mesoscale meteorological phenomena: Katabatic winds and boundary layer convection. *Planetary and Space Science*, 59, 915–922. https://doi.org/10.1016/j.pss.2010.04.025
- Spiga, A. (2019). The planetary boundary layer of Mars. In Oxford research encyclopedia of planetary science (p 118. 130). Oxford University Press. https://doi.org/10.1093/acrefore/9780190647926.013.130
- Spiga, A., Barth, E., Gu, Z., Hoffmann, F., Ito, J., Jemmett-Smith, B., et al. (2016). Large-eddy simulations of dust devils and convective vortices. Space Science Reviews, 203, 245–275. https://doi.org/10.1007/s11214-016-0284-x
- Spiga, A., Faure, J., Madeleine, J.-B., Määttänen, A., & Forget, F. (2013). Rocket dust storms and detached dust layers in the Martian atmosphere. Journal of Geophysical Research (Planets), 118, 746–767. https://doi.org/10.1002/jgre.20046
- Spiga, A., Forget, F., Lewis, S. R., & Hinson, D. P. (2010). Structure and dynamics of the convective boundary layer on mars as inferred from large-eddy simulations and remote-sensing measurements. *Quarterly Journal of the Royal Meteorological Society*, 136, 414–428. https:// doi.org/10.1002/qi.563
- Spiga, A., Hinson, D., Madeleine, J., Navarro, T., Millour, E., Forget, F., & Montmessin, F. (2017). Snow precipitation on Mars driven by cloud-induced nighttime convection. *Nature Geoscience*, 10, 652–657. https://doi.org/10.1038/ngeo3008
- Spiga, A., Murdoch, N., Lorenz, R., Forget, F., Newman, C., Rodriguez, S., & Banerdt, W. B. (2021). A study of daytime convective vortices and turbulence in the martian planetary boundary layer based on half-a-year of InSight atmospheric measurements and large-eddy simulations. *Journal of Geophysical Research: Planets*, 126(1), e2020JE006511. https://doi.org/10.1029/2020je006511
- Stanzel, C., Pätzold, M., Williams, D. A., Whelley, P. L., Greeley, R., Neukum, G., & HRSC Co-Investigator Team. (2008). Dust devil speeds, directions of motion and general characteristics observed by the Mars express high resolution stereo camera. *Icarus*, 197, 39–51. https:// doi.org/10.1016/j.icarus.2008.04.017
- Sutton, J. L., Leovy, C. B., & Tillman, J. E. (1978). Diurnal variations of the Martian surface layer meteorological parameters during the first 45 sols at two Viking lander sites. *Journal of the Atmospheric Sciences*, 35, 2346–2355. https://doi. org/10.1175/1520-0469(1978)035<2346:dvotms>2.0.co;2
- Toigo, A. D., Richardson, M. I., Ewald, S. P., & Gierasch, P. J. (2003). Numerical simulation of Martian dust devils. Journal of Geophysical Research (Planets), 108, 5047. https://doi.org/10.1029/2002JE002002
- Toigo, A. D., Richardson, M. I., Wang, H., Guzewich, S. D., & Newman, C. E. (2018). The cascade from local to global dust storms on Mars: Temporal and spatial thresholds on thermal and dynamical feedback. *Icarus*, 302, 514–536. https://doi.org/10.1016/j.icarus.2017.11.032
- Tyler, D., Barnes, J. R., & Skyllingstad, E. D. (2008). Mesoscale and large-eddy simulation model studies of the Martian atmosphere in support of Phoenix. *Journal of Geophysical Research (Planets)*, 113(E12), E00A12. https://doi.org/10.1029/2007JE003012
- Wu, Z., Richardson, M. I., Zhang, X., Cui, J., Heavens, N. G., Lee, C., et al. (2021). Large eddy simulations of the dusty martian convective boundary layer with MarsWRF. Journal of Geophysical Research: Planets, 126, e2020JE006752. https://doi.org/10.1029/2020JE006752